Author’s response to reviews

Title: Mesenteric cysts and mesenteric venous thrombosis led to intestinal necrosis in pregnancy and managed with laparotomy. Case report and review of the literature

Authors:

Aris Giannos (arisiannos@yahoo.gr)
Sofoklis Stavrou (sfstavrou@yahoo.com)
Nikolaos Goumalatsos (ngoum@yahoo.gr)
George Fragkoulidis (fragkgeo@yahoo.gr)
Chra Eleni (eleni.chra@yahoo.com)
Dimitrios Argiropoulos (argidimi1980@yahoo.it)
Peter Drakakis (pdrakakis@hotmail.com)
Dimitrios Loutradis (LOUTRADI@OTENET.GR)

Version: 2 Date: 03 May 2017

Author’s response to reviews:

Point-by-point response to reviewer:

1. Thank you for the comment. We added a relevant text (page 3, line 29 - page 4, line 1). We report the vital signs of the patient on admission (page 3, line 27-29).

2. Thank you for your comment. We divided the case report section into different paragraphs.